Author's response to reviews

Title: Synergizing Expectation and Execution for Stroke Communities of Practice Innovations

Authors:

Lise Poissant (lise.poissant@umontreal.ca)
Sara Ahmed (sara.ahmed@mcgill.ca)
Richard J Riopelle (richard.riopelle@mcgill.ca)
Annie Rochette (annie.rochette@umontreal.ca)
Helene Lefebvre (helene.lefebvre@umontreal.ca)
Deborah Radcliffe-Branch (deborah.radcliffe-branch@muhc.mcgill.ca)

Version: 4 Date: 21 May 2010

Author's response to reviews: see over
Montreal, May 21, 2010

To: Martin Eccles and Brian Mittman  
Editors-in-chief  
BMC Implementation Science

Object: Re-submission of manuscript 3831897573655485

From: Lise Poissant, PhD

Dear Dr. Eccles and Dr. Mittman,

It is with great pleasure that I would like to re-submit the manuscript entitled “Synergizing Expectation and Execution for Stroke Communities of Practice Innovations” as a study protocol.

I revised the content and removed files as requested. However, I do not understand the request made regarding authors names and consequently, apart from adding one middle initial to one author’s name, no changes have been made. I would appreciate if you could call me to explain any additional changes that remain to be made to this section.

I look forward to receiving your comments on our manuscript.

Sincerely,

Lise Poissant, PhD
Centre for Interdisciplinary Research in Rehabilitation of Greater Montreal  
Institute of Rehabilitation Gingras-Lindsay of Montreal, 6300 Darlington,  
School of Rehabilitation, University of Montreal,  
Montreal (Qc) Canada H3S 2J4  
Tel: (514) 340-2111 ext 2277  
Fax: (514) 340-2154

lise.poissant@umontreal.ca